 Vc/30-Aug-04
Week36 starting 30Aug  04

SM18 Operation Meeting (Tuesdays)

 Bench status  on Monday 30 aug04  ~17.40 hrs 

	TBA1
	MBAL3152
	Prep 0 Scan Bench
	299.1

	TBA2
	MBBL2093
	ICS 5 Traveller Last
	300.8

	TBB1
	SSS018
	Finished Ready to go Final scan
	299.2

	TBB2
	MBBL3196
	Finished Ready to go Final scan
	300.3

	TBC1
	MBAL3145
	PT 6.1 Training 1
	6.4

	TBC2
	MBBL1146
	Prep 5 Pumping + Cool Down
	1.9

	TBD1
	SSS044
	PT 12 Warm Up
	296.3

	TBD2
	SSS038
	Prep 5 Pumping + Cool Down
	87.8

	TBE1
	MBAL3169
	Prep 5 Pumping + Cool Down
	4.9

	TBE2
	MBBR1036
	PT 6.4 Training 4
	4.6

	TBF1
	MBAR1133
	Prep 1 Lyre Test
	298.5

	TBF2
	MBAR1140
	PT 12 Warm Up
	55.4


8/30/2004 5:42:23 PM    Temperature data are taken from TBSM18 site
Coordinator this week: Marco; MTM support on call :  (16-3602).

11 Magnets [ 10  dipoles( 0 with MM)+ 1 SSS]   Disconnected after tests last week ( Sat Midnight to Sat midnight)  
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.
· ( SSS work : New issues
CRYO PRIORITY SETTINGS & PAPER Sheet for OP Crew
Results by Week for PREVIOUS week 

	Magnets under Test from 22-08-2004 to 28-08-2004 

	-
	Bench
	Magnet
	Shafts
	Start Date .
	Connect
Time
	Cooling 
Time
	Cold Test 
Time
	Warming 
Time
	Disconnect
Time
	End Date
	Total
	
	

	1
	TBA1
	MBBL2105 
	[image: image1.png]



	16-08-2004 
	78 h.
	32 h.
	19 h.
	21 h.
	6 h.
	23-08-2004 
	168 hours
	
	

	2
	TBA2
	MBBL3102 
	[image: image2.png]



	19-08-2004 
	30 h.
	72 h.
	29 h.
	18 h.
	10 h.
	25-08-2004 
	144 hours
	
	

	3
	TBB2
	MBAL3186 
	[image: image3.png]



	18-08-2004 
	30 h.
	49 h.
	50 h.
	11 h.
	8 h.
	24-08-2004 
	144 hours
	
	

	4
	TBC1
	MBBL2104 
	[image: image4.png]



	18-08-2004 
	27 h.
	74 h.
	19 h.
	37 h.
	16 h.
	25-08-2004 
	168 hours
	
	

	5
	TBC2
	MBBL1143 
	[image: image5.png]



	21-08-2004 
	41 h.
	25 h.
	57 h.
	15 h.
	6 h.
	27-08-2004 
	144 hours
	
	

	6
	TBD2
	SSS012 
	[image: image6.png]



	09-08-2004 
	132 h.
	66 h.
	130 h.
	16 h.
	9 h.
	24-08-2004 
	360 hours
	
	

	7
	TBE1
	MBAL3173 
	[image: image7.png]



	17-08-2004 
	22 h.
	49 h.
	23 h.
	31 h.
	15 h.
	23-08-2004 
	144 hours
	
	

	8
	TBE1
	MBAL3201 
	[image: image8.png]



	23-08-2004 
	27 h.
	25 h.
	38 h.
	30 h.
	7 h.
	28-08-2004 
	120 hours
	
	

	9
	TBE2
	MBBL3197 
	[image: image9.png]



	21-08-2004 
	40 h.
	35 h.
	27 h.
	23 h.
	5 h.
	26-08-2004 
	120 hours
	
	

	10
	TBF1
	MBBL3136 
	[image: image10.png]



	13-08-2004 
	42 h.
	107 h.
	38 h.
	18 h.
	7 h.
	22-08-2004 
	216 hours
	
	

	11
	TBF2
	MBBL2084 
	[image: image11.png]



	19-08-2004 
	39 h.
	45 h.
	20 h.
	26 h.
	7 h.
	24-08-2004 
	120 hours
	
	


Quench Information
	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBBL2105
	10720 A.
(D2-U)
	12221 A.
(D2_L)
	[image: image12.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBBL3102
	12098 A.
(D2-L)
	27 A.
(D1_L)
	12850 A.
(No Data)
	[image: image13.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	MBAL3186
	12327 A.
(D1-U)
	12468 A.
(D1_U)
	[image: image14.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	MBBL2104
	11442 A.
(D1-L)
	12642 A.
(D2-U)
	[image: image15.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	MBBL1143
	11242 A.
(D1_U)
	11711 A.
(D1_L)
	11666 A.
(D1-L)
	11786 A.
(D1 L)
	11808 A.
(D1 L)
	12003 A.
(D1_L)
	12065 A.
(D1_L)
	12178 A.
(D2-U)
	12228 A.
(D1_L)
	0 A.
(D1_L)
	[image: image16.png]



	-
	-

	6
	SSS012
	11592 A.
(QI-34)
	12371 A.
(QE-34)
	12552 A.
(QI-12)
	12626 A.
(QE-12)
	12762 A.
(QE-12)
	12851 A.
(No Data)
	12851 A.
[image: image17.jpg]



	-
	-
	-
	-
	-
	-

	7
	MBAL3173
	11277 A.
(D1-U)
	11277 A.
(D1_U)
	12850 A.
[image: image18.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	MBAL3201
	12232 A.
(D1_L)
	12312 A.
(D2-L)
	[image: image19.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	MBBL3197
	11532 A.
(D2_U)
	12687.5 A.
(D2-L)
	[image: image20.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	10
	MBBL3136
	11286 A.
(D1-L-S13)
	11447 A.
(D2-U-S13)
	11789 A.
(D2-CFB)
	11977 A.
(D1-CFB)
	12089 A.
(D2-CFB)
	[image: image21.png]



	-
	-
	-
	-
	-
	-
	-

	11
	MBBL2084
	11922 A.
(D2-U)
	12687 A.
(D2-L)
	[image: image22.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-


Bench status  and tests for now( Monday 30 Aug/at 17.45 hrs)  & next few days :  see Video Display

	Magnets under Test on 30-08 5:44:28 PM 

	-
	Bench
	Magnet
	Shafts
	Start
	Connect
Time
	Cooling 
Time
	Cold Test 
Time
	Warming 
Time
	Disconnect
Time
	Last Test
Desc
	
Time
	
I
	
Temp.
	Total

	1
	TBA1
	MBAL3152 
	[image: image23.png]



	30-08 
	6 h.[image: image24.png]



	-
	-
	-
	-
	Prep 0 Scan Bench
	30-08
(13 : 20) 
	0 A. 
	300 K
	6 hours

	2
	TBA2
	MBBL2093 
	[image: image25.png]



	26-08 
	30 h.
	27 h.
	27 h.
	26 h.
	2 h.
	ICS 5 Traveller Last
	30-08
(16 : 38) 
	0 A. 
	300 K
	113 hours

	3
	TBB1
	SSS018 
	[image: image26.png]



	17-08 
	81 h.
	45 h.
	85 h.
	7 h.
	73 h.
	Finished Ready to go Final scan
	29-08
(12 : 29) 
	0 A. 
	300 K
	320 hours

	4
	TBB2
	MBBL3196 
	[image: image27.png]



	24-08 
	44 h.
	23 h.
	50 h.
	19 h.
	7 h.
	Finished Ready to go Final scan
	30-08
(11 : 08) 
	0 A. 
	300 K
	149 hours

	5
	TBC1
	MBAL3145 
	[image: image28.png]



	25-08 
	122 h.[image: image29.png]



	-
	-
	-
	-
	PT 6.1 Training 1
	30-08
(11 : 55) 
	10959 A. 
	1.9 K
	122 hours

	6
	TBC2
	MBBL1146 
	[image: image30.png]



	27-08 
	23 h.
	46 h.[image: image31.png]



	-
	-
	-
	Prep 5 Pumping + Cool Down
	28-08
(19 : 08) 
	0 A. 
	Cooling
	69 hours

	7
	TBD1
	SSS044 
	[image: image32.png]



	18-08 
	152 h.
	47 h.
	83 h.
	6 h.[image: image33.png]



	-
	PT 12 Warm Up
	30-08
(11 : 46) 
	0 A. 
	Warming
	288 hours

	8
	TBD2
	SSS038 
	[image: image34.png]



	24-08 
	130 h.
	19 h.[image: image35.png]



	-
	-
	-
	Prep 5 Pumping + Cool Down
	29-08
(22 : 00) 
	0 A. 
	Cooling
	149 hours

	9
	TBE1
	MBAL3169 
	[image: image36.png]



	28-08 
	26 h.
	23 h.[image: image37.png]



	-
	-
	-
	Prep 5 Pumping + Cool Down
	29-08
(18 : 02) 
	0 A. 
	Cooling
	49 hours

	10
	TBE2
	MBBR1036 
	[image: image38.png]



	26-08 
	46 h.
	29 h.
	30 h.[image: image39.png]



	-
	-
	PT 6.4 Training 4
	30-08
(13 : 33) 
	12170 A. 
	1.9 K
	105 hours

	11
	TBF1
	MBAR1133 
	[image: image40.png]



	29-08 
	22 h.[image: image41.png]



	-
	-
	-
	-
	Prep 1 Lyre Test
	30-08
(03 : 11) 
	0 A. 
	300 K
	22 hours

	12
	TBF2
	MBAR1140 
	[image: image42.png]



	24-08 
	38 h.
	6 h.
	91 h.
	2 h.[image: image43.png]



	-
	PT 12 Warm Up
	30-08
(15 : 29) 
	0 A. 
	Warming
	137 hours


Quench Information
	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBAL3152
	[image: image44.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBBL2093
	11123 Amp.
(D1-U)
	12358 Amp.
(D2-L)
	[image: image45.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	SSS018
	11295 Amp.
(QI_12)
	12331 Amp.
(QE_12)
	[image: image46.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	MBBL3196
	9805 Amp.
(D2_L)
	11857 Amp.
(D2_L)
	12850 Amp.
([image: image47.jpg]


)
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	MBAL3145
	10959 Amp.
(D1_U_S01)
	[image: image48.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	MBBL1146
	[image: image49.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	SSS044
	12067.3 Amp.
(QE-34)
	12856 Amp.
(No Data)
	[image: image50.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	SSS038
	[image: image51.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	MBAL3169
	[image: image52.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	10
	MBBR1036
	11825 Amp.
(D1-L)
	12213 Amp.
(D1-L)
	11950 Amp.
(D1_L)
	12170 Amp.
(D1-L)
	[image: image53.jpg]



	-
	-
	-
	-
	-
	-
	-
	-

	11
	MBAR1133
	[image: image54.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	12
	MBAR1140
	12465 Amp.
(D1_U, S1)
	12666 Amp.
(D1_L-S13)
	12811 Amp.
(No Data)
	[image: image55.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-


General Problems last week :  
	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday_24_N
	TBB1
	IAP @ warm on B2: B1 is called. Need to discharge heaters but Heaters 1 &2 did not discharge.
	MTM
	
	Salem called.

	Wednesday_25_N


	TBA2
	Searching for 2093 magnet in MTF two names appeared HCLBBL2093 and HCLABL2093
	TS
	MTF
	EDMS support informed by Stephane. Correction on going.

	Wednesday_25_N


	TBB1
	Warm-up quench repeated because information not given to ACR team (No SMS, no verbal information)
	MTM
	
	

	Wednesday_25_N


	TBB
	Still no hf and lf status windows to display the magnet names in TEMA
	AB-CO
	
	

	Wednesday_25_N


	TBB1
	During warm –up quench  of SSS dipole pens selected instead of SSS pens
	MTM
	
	

	Thursday _26_A
	TBD2
	Survey on SSS038: magnet moving during cool down.
	MTM
	
	Survey repeated?

	Thursday _26_A
	TBD1
	Rotation of shaft in aperture 1 not good (4% RMS)
	MTM
	
	Pierre informed Stephane and P.Legrand. Problem in the coupler of the TRU. Solved.

	Thursday _26_N


	TBB1
	Roll angle on SSS018 out of limit : 23 mrad measurements repeated several times.


	MTM
	
	

	Thursday _26_N


	TBE2
	HV final at warm (WP07) not available in MTF.ICS informed that they have done the test and that it was OK. 
	ICS
	
	File has to be uploaded in MTF.

	Friday_27_M
	TBD1
	SPA test: Water fault.
	AB/PO
	
	Salem informed. This persists even then water flow OK. AB/PO informed. They solved the problem.

	Friday_27_M
	TBD1
	SPA test: Communication problem. Unable to launch from TEMA
	AB/CO
	TEMA
	Salem solved the problem by quitting and resetting TEMA.

	Friday_27_M
	TBD1
	IAP:  T coil measurements: Voltage limit setting too low (52 V)
	MTM
	IAP
	Salem increased the limit to 65 V..

	Friday_27_M
	TBB2
	Rotary switch blocked in a position between CC and B2
	MTM
	
	Salem switched off the controller and repeat the operation.

	Friday_27_M
	TBD1
	Quench signals too noisy?
	MTM
	
	Solved by Salem.

	Friday_27_M
	TBD2
	Wire broke during rot wire measurements.
	MTM
	
	Following the list proposed last week : XY repeated then rotwire again.

	Friday_27_A
	TBB1
	SSS018 : Problem with hf/lf DAQ : Wrong magnet setting.

Dipole setting launched instead of quadrupole one.
	MTM
	
	

	Friday_27_A
	TBD1
	Unable to find MM position.
	MTM
	
	We have to look into b2 value and not b1.

	Friday_27_N
	TBA1
	Training 2 of 3199: TEMA hung. TEMA off and re-launched.

After the quench : AQA could not analyse the quench

Alarm : All 4 heaters should fire after  trigger
	AB/CO
	TEMA
	Analysis done locally.

TC informed : give the OK to continue.

	Friday_27_N

Saturday_28_M


	TBD2
	Rotwire MQT in Ap1 not  performed : PC blocked.

SW15 re-launched : wire broke.


	MTM
	
	Salem shut down the station and restarted it.

	Saturday_28_M
	TBB2
	SPA not getting launched. PC fault . Interlock could not be reset
	AB/CO?
	
	Salem solved the problem by resetting the PLC.

	Saturday_28_M
	
	Hf/lf viewer not working properly in the control room.
	AB/CO
	Hf/lf
	Salem advised to work in local stations..

	Saturday_28_A
	TBA2
	HWREC blocked on test of magnet 2093
	AB/CO
	TEMA
	Salem switched off TEMA and it worked again.

	Saturday_28_A
	TBD1
	Rotwire MQT: Serial port problem with power application.

Then during the measurements wire broke.

Inclinometer not working correctly.
	AB/CO
	
	

	Saturday_28_N
	TBF2
	Current in PC drops during LHC cycle
	?
	
	Min NRJ and cycle repeated.

	Saturday_28_N
	TBB2
	Water interlock in PC
	AB/PO
	
	Called Piquet AB/PO: he increased the level of water.

	Sunday_29_M
	TBF2
	No quench data in DAQ disk: wrong magnet name: I057 instead of F140.
	MTM
	
	Field have to be transferred Monday.

	Sunday_29_M

Sunady_29_N
	TBF2
	MMP problem: Trigger not received. Switch off/on the crates: error remains
	AB/CO.
	MMP
	Salem start the investigation. Helped by Alessandro Sunday evening. Crates, cable changed. Still error. Change the sun station with sun MTA34 then OK.

Communication card in the Sun not working?

H.Reymond will investigate the problem Monday.



	Sunday_29_M
	TBF2
	Hfoff slow on F cluster. No analysis of quench possible on control room
	AB/CO
	hf
	Salem found that analysis is possible locally. No data for quench antenna.

	Sunday_29_A
	TBC1
	Amplifier problem 
	AB/CO
	
	Salem solved the problem by changing RS232 converter for amplifier 2 in MM mobile rack

	Sunday_29_A
	TBD1
	Slow PA when ramping for quench at 11850 A.
	
	
	

	Sunday_29_N
	TBE2
	AQA not working properly in TEMA and local server.
	AB/CO
	AQA
	Analysis done with files DAQ disk 2. Informed AB/CO Monday.

	Sunday_29_N
	TBC1
	Wrong name on active bench : launching of SPA test not possible .
	MTM
	
	Salem launched it.

	Sunday_29_N
	TBC1
	MM1: Error in amplifier communication
	AB/CO
	
	Salem solved the problem: changed of the amplifier card?

	Sunday_29_N
	TBD1
	Corrector power test , GH circuit. TEMA blocked
	AB/CO
	TEMA
	Salem informed. Emergency stop pushed. Test re-launched locally.

	Sunday_29_N
	TBC1
	AQA not working from control room.
	AB/CO
	AQA
	Salem advised to do the analysis through the local server.

	Sunday_29_N
	TBE1
	SPA test. PLC fault : cryo interlock 
	ACR
	
	








